Author’s response to reviews

Title: The typically developing paediatric foot: how flat should it be? A systematic review

Authors:

Hayley Uden (hayley.uden@unisa.edu.au)

Rolf Scharfbillig (rolf.scharfbillig@unisa.edu.au)

Ryan Causby (ryan.causby@unisa.edu.au)

Version: 2 Date: 07 Jul 2017

Author’s response to reviews:

Thank you for your time, I have completed all of the below, I hope I havn't missed anything,
Kind regards, Hayley

1. Please change title to: The typically developing paediatric foot: how flat should it be? A systematic review. (i.e. replaced semi-colon with colon). Please ensure this change is made to both the manuscript file and the online submission system.

   Done.

2. Please remove Dr titles from authors’ names

   Done.

3. Please add city, postcode and country to author affiliations.

   Done.

4. Abstract: please remove all capitals from subheadings, i.e. BACKGROUND should be Background, followed by a colon (:).

   Done.

5. Remove quotation marks around ‘how flat’ throughout manuscript.
6. The conclusion section of the abstract is still far too long and reads more like a discussion. Please reduce to a maximum of 3 sentences.
Done.

7. Main text of manuscript: Replace all capitals from all subheadings, i.e: BACKGROUND should be Background.
Done.

8. Remove unnecessary capital letters from subheadings, i.e: Study Selection should be Study selection, Statistical Methods should be Statistical methods.
Done.

9. Remove all “insert figure/table here” sections as these are not required.
Done.

10. Please run a spell-check on the entire document. - Nothing came up, is there somewhere in particular I have made a mistake?
Done. Nothing came up, is there somewhere in particular that I have missed?

11. Conclusion: far too long, and some sections read like a discussion. Reduce to one paragraph.
Done.

12. The references are formatted incorrectly. Please check the guidelines and amend, ensuring that the journal names are correctly abbreviated.
Done.
13. Table: format references as Bertsch et al [x] rather than (Bertsch, Unger, et al. 2004), using the same reference numbers as the reference list.

Done.